Author’s response to reviews

Title: A rare case of anti-LGI1 limbic encephalitis with concomitant positive NMDAR antibodies

Authors:

Tuo Ji (juitar_hope@126.com)

Zhi Huang (zzhuangzhi@163.com)

Yajun Lian (liyajun369@yeah.net)

Chengze Wang (supermaorios@163.com)

Qiaoman Zhang (1318696664@qq.com)

Jinghong Li (lijinghong123@163.com)

Version: 1 Date: 28 Aug 2020

Author’s response to reviews:

Dear Prof Coore,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate you and the reviewers very much for your positive and constructive comments and suggestions on our manuscript entitled “A rare case of anti-LGI1 limbic encephalitis with concomitant positive NMDAR antibodies” (ID: NURL-D-20-00949).

We have studied your kindly comments carefully and tried our best to revise our manuscript, which we hope will meet with approval. According to the comments, we deleted the textual overlap with other previously published works and rephrased this section (Discussion section, paragraph 3, line 21-25, page 5). Besides, we removed the highlights and uniformed the text in different colours. Attached please find the revised manuscript as a single, final, clean version, which we would like to submit for your kind consideration.

We would like to express our great appreciation to you and reviewers for comments on our paper. Looking forward to hearing from you.

Thank you and best regards.
Yours sincerely,

Yajun Lian